Author’s response to reviews

Title: Krüppel-like factor 4 expression in oral carcinoma cells and hypermethylation at the gene promoter

Authors:

Ayumi Yamaguchi (ayumi.y.ii.may.seven1@gmail.com)
Karen Kuroyama (kyun1096@yahoo.co.jp)
Ayana Tokura (rococo15milktea@gmail.com)
Atsushi Saito (masuzushimango@yahoo.co.jp)
Huhga Arikawa (americanfootboll4223@yahoo.co.jp)
Takahisa Hasebe (soccer.hasebe@gmail.com)
Dai Usui (daiusui1992@yahoo.co.jp)
Kasuke Yamaguchi (kyama@tky.ndu.ac.jp)
Tadashige Chiba (tchiba@tky.ndu.ac.jp)
Kazushi Imai (kimai@tky.ndu.ac.jp)

Version: 2 Date: 24 Jan 2016

Author’s response to reviews:

Thank you very much for your e-mail.

We used established cells lines that were commercially available from cell bank, and did not use any cells that are required to conform the Helsinki Declaration in the study (OHEA-D-15-00150R1).

Although you mentioned that we need to declare the ethical approval from our institutional commitment, I am not sure still whether we need to declare it in the manuscript.

Of course we have an approval to use recombinant DNA in the study.

I removed all track changes in the revised manuscript.